Author’s response to reviews

Title: Repeated electroacupuncture treatment attenuated hyperalgesia through suppression of spinal glial activation in chronic neuropathic pain rats

Authors:
junying wang (wjyanguning@aliyun.com)
yonghui Gao (reneversun@163.com)
lina qiao (qiaolina2088@163.com)
jianliang zhang (drzhangjl@yahoo.com)
Chenglin Duanmu (2360663811@qq.com)
Yaxia Yan (515109759@qq.com)
Shuping Chen (809107218@qq.com)
junling liu (13521898023@163.com)

Version: 2 Date: 01 Jan 2018

Author’s response to reviews:

Hsiu-Mei Chiang (Reviewer 2): Authors have revised the manuscript according to the comments; however, there are still some grammar and type errors. Please revise the manuscript thoroughly.

Many thanks to the reviewer’s criticism. We corrected some wrong words and sentences in the revised manuscript.

We also revised the figures. In the figures, “The ipsilateral side” is changed to “Ipsi.”; ”The contralateral side” is changed to “Cont.”; ”the number of Iba...”is changed to “number of ..”; “the meanintensity of GFAP” is changed to “meanintensity of GFAP”.